Author’s response to reviews

Title: Incidence of obstetrical thrombotic thrombocytopenic purpura in a retrospective study within thrombocytopenic pregnant women. A difficult diagnosis and a treatable disease

Authors:

Yahsou Delmas (yahsou.delmas@chu-bordeaux.fr)
Sébastien Hélou (helou@clinique-delay.fr)
Pierre Chabanier (pierre.chabanier@chu-bordeaux.fr)
Anne Ryman (anne.ryman@chu-bordeaux.fr)
Fanny Pelluard (fanny.pelluard@chu-bordeaux.fr)
Dominique Carles (dominique.carles@u-bordeaux.fr)
Pierre Boisseau (pierre.boisseau@chu-nantes.fr)
Agnès Veyradier (agnes.veyradier@lrb.aphp.fr)
Jacques Horovitz (jacques.horovitz@chu-bordeaux.fr)
Paul Coppo (paul.coppo@sat.aphp.fr)
Christian Combe (christian.combe@chu-bordeaux.fr)

Version: 5 Date: 4 May 2015

Author’s response to reviews:

Dear Editor,

I can attest today that I have received the written informed consent of all 4 ADAMTS13 deficient patients for the publication of their individual clinical details. This is stated in the manuscript page 8 lines 189-190.

Kind regards
Dr Yahsou Delmas